Massachusetts Institute of Technology Handout 4 6.042J/18.062J: Mathematics for Computer Science February 1, 1994 Professors Charles E. Leiserson and Tom Leighton

Recitations and Tutorials

The following is the arrangement of recitations and tutorials.

Section Recitation (Friday) Tutorial (Monday) 1 1:00, 26{210 1:00, 24{320 2 2:00, 38{368 3 Steve Ponzio 3:00, 24{316 4 4:00, 24{316 5 2:00, 26{210 11:00, 34{498 6 1:00, 34{498 7 Ravi Sundaram 2:00, 34{498 8 3:00, 34{498 9 3:00, 34{303 1:00, 38{495 10 2:00, 38{495 11 Brian Taylor 3:00, 38{495 12 4:00, 38{495